Author’s response to reviews

Title: Perceptions about trust: A phenomenographic study of clinical supervisors in occupational therapy

Authors:

Pernilla Lundh (pernilla.lundh@ki.se)
Per Palmgren (per.palmgren@ki.se)
Terese Stenfors (terese.stenfors@ki.se)

Version: 3 Date: 18 Oct 2019

Author’s response to reviews:

We hereby submit our clean version of the revised paper “Perceptions about trust: A phenomenographic study of clinical supervisors in occupational therapy”.

Comments by editor

Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Response by the authors and revisions made

We use no abbreviations in our text except authors’ initials and authors’ academic credits. These are normally not reported as abbreviations.

Comments by editor

In order to be in line with journal requirements, please include the following headings within your manuscript: Background, Methods, Results, Discussion, Conclusions

Response by the authors and revisions made

Required headings have been amended accordingly.
Comments by editor

Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response by the authors and revisions made

Responses to editor have been inserted correctly in the Reviewers box in the editorial system.

A clean final version of the manuscript has been uploaded.

Updated tables have been uploaded separately.